Author's response to reviews

Title: Association between socio-economic status and hemoglobin A1c levels in a Canadian primary care adult population without diabetes

Authors:

Babak Aliarzadeh (babaka@cpcssn.org)
Michelle Greiver (mgreiver@rogers.com)
Kieran M Moore (kieran.moore@kflapublichealth.ca)
Paul Belanger (paul.belanger@kflapublichealth.ca)
Rahim Moineddin (rahim.moineddin@utoronto.ca)
Christopher Meaney (christopher.meaney@utoronto.ca)
David White (david.white@utoronto.ca)
Ambreen Moazzam (ambreenm@cpcssn.org)

Version: 3 Date: 30 November 2013

Author's response to reviews: see over
November 27th 2013

To: The Editor,

BioMed Central

Dear Mr. Christos Lionis

Thank you for the opportunity to respond to the reviewers’ comments. Please find our responses below.

1. To avoid the term "diabetic" and to replace with the term "patients with diabetes.

The term “non-diabetic” was replaced with “persons without diabetes” in the manuscript.

2. To comment in the discussion section and their conclusion the external validity of their study's findings. In more simple words, to respond to what extent their findings could be generalized in other clinical settings outside of North America.

A paragraph was added to the end of the discussion section to address this recommendation.

We appreciate this second opportunity to respond to the reviewers’ comments and to resubmit a revised version of the paper.